Author's response to reviews

Title: Postprandial energy metabolism and substrate oxidation in response to the inclusion of a sugar- or non-nutritive sweetened beverage with meals differing in protein content

Authors:
Shanon Casperson (shanon.casperson@ars.usda.gov)
Clint Hall (clint.hall@ars.usda.gov)
James Roemmich (james.roemmich@ars.usda.gov)

Version: 2 Date: 01 May 2017

Author’s response to reviews:
James Mockridge, PhD
Editor, BMC Nutrition

Dear Dr. Mockridge,

We are honored that you have accepted our manuscript titled “Postprandial energy metabolism and substrate oxidation in response to the inclusion of a sugar- or non-nutritive sweetened beverage with meals differing in protein content” for publication. As stipulated in your acceptance letter we have added the date of registration alongside the registration number in both the abstract and the manuscript body, and we have moved the embedded tables from the text and have placed them after the references.

Healthy Regards,

Shanon Casperson, PhD, DTR
Research Biologist
Grand Forks Human Nutrition Research Center
USDA-Agricultural Research Services
2420 2nd Ave. North
Grand Forks, ND 58203-9034
Phone: 701-795-8497
Fax: 701-795-8230
Email: shanon.casperson@ars.usda.gov